Author's response to reviews

Title: Diabetes Mellitus in North West Ethiopia: a community based study

Authors:

Solomon Mekonnen Abebe Mr (solomekonnen@yahoo.com)
Yemane Berhane Professor (yemanbeberhane@ethionet.et)
Alemayehu Worku Dr. (alemayehuwy@yahoo.com)
Abebayehu Assefa Dr. (abebayehua@et.afro.who.int)

Version: 3 Date: 16 January 2014

Author's response to reviews: see over
To: BMC Public Health Editor-in-Chief

Subject: - Point by point response for the comments

Title: Diabetes Mellitus in North West Ethiopia: a community based study

Reference: MS: 2415796310321136

Dear Editor-in-Chief

We are very grateful for the consideration of the manuscript. In accordance with the reviewers valuable comments and recommendations, we have revised the (MS 3) and we hereby submit the revised work for your consideration.

Reviewer 1

<table>
<thead>
<tr>
<th>Question No.</th>
<th>Question</th>
<th>Response</th>
</tr>
</thead>
<tbody>
<tr>
<td></td>
<td>Minor Essential revision Needs some language corrections before being Published</td>
<td>Further editorial corrections were made to improve the language.</td>
</tr>
</tbody>
</table>

Reviewer 2

<table>
<thead>
<tr>
<th>Question No.</th>
<th>Question</th>
<th>Response</th>
</tr>
</thead>
<tbody>
<tr>
<td></td>
<td>There are spelling mistakes in many places e.g. ‘opner’ in the last sentence in page 11. The second sentence in page 10, paragraph 3, need correction. The last sentence in the results should be moved 2 sentences up. Page 7, last sentence: please replace the word ‘is’ with ‘was’.</td>
<td>Correction made as suggested</td>
</tr>
<tr>
<td></td>
<td>Needs some language corrections before being published</td>
<td>We have obtained further editorial assistance to improve the language (native English speaking colleague)</td>
</tr>
</tbody>
</table>

With Best Regard!!

The Authors